Title: PRIME - PRocess modelling in ImpleMEntation research: selecting a theoretical basis for interventions to change clinical practice.

Authors:

Dr Anne Walker (a.e.walker@abdn.ac.uk)
Prof Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Marie Johnston (m.johnston@abdn.ac.uk)
Nigel Pitts (n.b.pitts@dundee.ac.uk)
Nick Steen (nick.steen@ncl.ac.uk)
Prof Martin P Eccles (martin.eccles@ncl.ac.uk)

Version: Date: 18 Dec 2003

Dear Editor,
As requested in your email I have structured the abstract (Background and Methods - discussion was not appropriate.
I have removed the two instances of the text "Insert Table n here".
I have formatted the box as a table and altered the text to refer to Table 2.
Yours sincerely,
Martin Eccles.